Author's response to reviews

Title: Injury Rates and Injury Risk Factors Among Federal Bureau of Investigation New Agent Trainees

Authors:

Joseph J Knapik (joseph.knapik@apg.amedd.army.mil)
Tyson Grier (tyson.grier@us.army.mil)
Anita Spiess (anita.spiess@us.army.mil)
David I Swedler (dswedler@jhsph.edu)
Keith G Hauret (keith.hauret@us.army.mil)
Bria Graham (bria.graham@us.army.mil)
James Yoder (james.yoder@ic.fbi.gov)
Bruce H Jones (bruce.h.jones@us.army.mil)

Version: 3 Date: 13 November 2011

Author's response to reviews: see over
Dear Sir or Madam,

We would like to thank the Associate Editor for finding the oversight in our manuscript entitled “Injury Rates and Injury Risk Factors among Federal Bureau of Investigation New Agent Trainees” (Manuscript Number 8319388905508038) which we submitted for consideration of publication in *BMC Public Health*. As the Associate Editor indicated, we have now changed average±SD to 'mean±SD'. We have also assured that the paper confirms to the journal style.

Please continue to direct correspondence regarding this manuscript to me as follows:
Dr Joseph Knapik
US Army Institute of Public Health
ATTN: MCHB-IP-DI
Directorate of Epidemiology and Disease Surveillance
Aberdeen Proving Ground (Edgewood Area), MD 21010
My phone number is (410) 436-1328, my FAX number is (410) 436-5449, and my e-mail address is joseph.knapik@us.army.mil.

We continue to look forward to the possibility of having our work published in *BMC Public Health* and await your final decision on our work.

Sincerely,